Author's response to reviews

Title: A factorial randomized controlled trial to evaluate the effect of micronutrients supplementation and regular aerobic exercise on maternal endothelium-dependent vasodilatation and oxidative stress of the newborn

Authors:

Ana C Aguilar de Plata (caplataa@yahoo.es)  
Julio C Mateus (icmateus@fundacionfes.org)  
Robinson Ramirez-Velez (robin640@hotmail.com)  
Isabella Echeverry (isabella_ech@hotmail.com)  
Jose G Ortega (guileort@yahoo.com)  
Mildrey Mosquera (mildrey@yahoo.com)  
Blanca Salazar (bsblasalaz@gmail.com)  
Sandra L Girón (sandragiron@fundacionfes.org)  
Wilmar Saldarriaga (wsaldarriaga0608@yahoo.com)  
Miryam Romero (miryamrom@gmail.com)

Version: 5 Date: 12 January 2011

Author's response to reviews:

Santiago de Cali, January 12 2011

Professor
David Moher
Editors
Trials

RE: MS: 1488023423382489 “A factorial randomized controlled trial to evaluate the effect of micronutrients supplementation and regular aerobic exercise on maternal endothelium-dependent vasodilatation and oxidative stress of the newborn”

Dear Editor,

Please find attached the revised version (v-4) of the manuscript “A factorial randomized controlled trial to evaluate the effect of micronutrients supplementation and regular aerobic exercise on maternal endothelium-dependent vasodilatation and oxidative stress of the newborn”. All authors have read and approved the paper, and declare that there is no conflict of interest with the subject of the manuscript. This manuscript was checked over by a native English speaker and the language.

Corresponding author 1
Ana Cecilia Aguilar de Plata
Calle 4B 36-00, Edificio 116, Quinto piso, Oficina 5502, Cali, Colombia
Tel.: +57 2 5185603
caplataa@yahoo.es

Corresponding author 2
Robinson Ramírez Velez
Calle 4B 36-00, Edificio 116, Quinto piso, Oficina 5502, Cali, Colombia
Tel.: +57 2 5185603
robin640@hotmail.com